Author's response to reviews

Title: Abernethy malformation: a case report

Authors:

Ashish Pathak (drashishp@rediffmail.com)
Nitin Agarwal (drnitinagrawal@rediffmail.com)
Jagdish Mandliya (drjagdish1@yahoo.co.in)
Prateek Gehlot (prateekgehlot@yahoo.com)
Mamta Dhaneria (drspdhaneria@rediffmail.com)

Version: 6 Date: 23 March 2012

Author's response to reviews: see over
To,

The Editor,

Title: Abernethy malformation: a case report

Revision 3

Dear Editor,

We have now included the consent statement in the methods section of the manuscript on page 7 as follows:

**Consent**

Written informed consent was obtained from the patient’s father for publication of this case report and accompanying images.

We hope that the above reply satisfies editor and the manuscript is accepted for publication at the earliest.

Thanking you in anticipation.

Ashish Pathak

Associate Professor,

Department of Pediatrics,

RD Gardi Medical College,

Ujjain